Author's response to reviews

Title: Study of the serotonin transporter (SLC6A4) and BDNF genes in French patients with non syndromic mental deficiency

Authors:

Refaat Rabagh (refaat_tb@yahoo.com)
Christian R Andres (andres@med.univ-tours.fr)
Sylviane Vedrine (sylviane.vedrine@univ-tours.fr)
Catherine Cherpi-Antar (cantar1@clubinternet.fr)
Rose-Anne thepault (thepault@med.univ-tours.fr)
Laurence Mignon (lmignon@neiglobal.com)
Diane Dufour-Rainfray (ddufourrainfray@yahoo.fr)
Claude Moraine (moraine@gmail.com)
Patrick Vourc'h (vourch@med.univ-tours.fr)

Version: 3 Date: 8 December 2009

Author's response to reviews: see over
Dear Editor,

Please find attached our revised manuscript entitled “Study of the serotonin transporter (SLC6A4) and BDNF genes in French patients with non syndromic mental deficiency” by Tabagh et al. for your consideration for BMC Medical Genetics. We report in this manuscript the first genotyping study on the SLC6A4 gene encoding the serotonin transporter 5-HTT and the BDNF gene in a population of patients with non syndromic mental deficiency.

As requested, we asked a native English speaking person (Dr L. Mignon, Carlsbad, CA, USA) to improve the manuscript.

Thank you for your consideration of our manuscript.

Sincerely yours,

Patrick VOURC'H, PhD
INSERM U930
Laboratoire de Biochimie et Biologie Moléculaire,
Hôpital Bretonneau, 2 Bd Tonnelle, 37044 Tours, France
tel: 02 34 37 89 10, fax: 02 47 47 86 13; vourch@med.univ-tours.fr